\begin{table}%t4
%\centering
\par
\caption{The sample of 170 BL Lac objects selected from \cite{nieppola2006}.}\label{tab:core} 
\begin{tabular}{llllllllllll}
\hline\hline\noalign{\smallskip}
IAU name&Source&$z$ & $\log \rm \nu_{peak}'$&$\log P_{\rm 408~ M}$  &$\log P_{\rm core}$
&$f_{\rm ext}$&Refs.&$\delta$&$\theta_{3}$&$\theta_{5}$ &$\theta_{10}$\\
\hline
 0006-063  &  NRAO 5             &      0.347  & 12.10  &  26.94$^{a}$ &     26.65   &    ...        &   1 &    6.0 &    0.0&  9.4  &   8.8      \\
0007+472  &  RX J0007.9+4711    &      0.302* & 15.81  &  25.63$^{b}$ &     25.18   &    ...        &   2 &    2.8 &   21.1& 18.7  &  14.3      \\
0035+598  &  1ES 0033+595       &      0.086  & 18.60  &  24.01$^{c}$ &     24.02   &    4.50       &   3 &    2.3 &   24.7& 21.1  &  15.9      \\
0040+408  &  1ES 0037+405       &      0.271* & 16.62  &  24.93$^{c}$ &     24.42   &     3.07      &   3 &    1.9 &   28.8& 23.9  &  17.7      \\
0050-094  &  PKS 0048-097       &      0.537  & 12.94  &  27.29$^{c}$ &     26.60   &  139.00       &   4 &    4.4 &   11.5& 13.1  &  10.9      \\
0110+418  &  NPM1G +41.0022     &      0.096  & 17.72  &  24.41$^{d}$ &     23.58   &    ...        &   2 &    1.1 &   43.5& 34.2  &  24.5      \\
0112+227  &  S2 0109+22         &      0.473* & 12.83  &  26.36$^{b}$ &     26.60   &     ...       &   2 &    8.5 &    0.0&  4.7  &   6.7      \\
0115+253  &  RXS J0115.7+2519   &      0.350  & 13.15  &  25.31$^{d}$ &     24.92   &    ...        &   2 &    2.6 &   22.5& 19.7  &  14.9      \\
0123+343  &  1ES 0120+340       &      0.272  & 17.96  &  24.86$^{c}$ &     24.75   &    2.59       &   3 &    2.9 &   19.9& 18.0  &  13.9     \\
0124+093  &  MS 0122.1+0903     &      0.339  & 15.37  &  23.58$^{c}$ &     23.60   &    0.08       &   5 &    2.0 &   28.4& 23.6  &  17.5      \\
0136+391  &  B3 0133+388        &      0.271* & 16.31  &  25.44$^{a}$ &     24.95   &     ...       &   2 &    2.4 &   23.7& 20.5  &  15.5      \\
0141-094  &  PKS 0139-09        &      0.733  & 12.77  &  27.17$^{c}$ &     26.99   &   50.00       &   6 &    7.4 &    0.0&  6.7  &   7.5      \\
0148+140  &  1ES 0145+138       &      0.125  & 15.76  &  24.22$^{e}$ &     22.88   &    ...        &   7 &    0.5 &   65.8& 49.9  &  34.9      \\
0153+712  &  8C 0149+710        &      0.022  & 14.65  &  23.99$^{b}$ &     23.50   &    ...        &   2 &    1.3 &   38.2& 30.4  &  22.0      \\
0201+005  &  MS 0158.5+0019     &      0.299  & 17.62  &  24.46$^{c}$ &     24.29   &    0.83       &   5 &    2.3 &   24.9& 21.3  &  16.0      \\
0208+353  &  MS 0205.7+3509     &      0.318  & 14.90  &  23.91$^{c}$ &     24.10   &    0.20       &   5 &    2.7 &   21.2& 18.8  &  14.4      \\
0214+517  &  87GB 02109+5130    &      0.049  & 17.53  &  24.50$^{b}$ &     23.94   &    ...        &   2 &    1.5 &   34.4& 27.8  &  20.3      \\
0222+430  &  3C 66A             &      0.440  & 15.20  &  27.55$^{a}$ &     26.65   &    ...        &   8 &    3.9 &   14.3& 14.6  &  11.8      \\
0232+202  &  1ES 0229+200       &      0.140  & 19.22  &  24.75$^{d}$ &     24.32   &    ...        &   7 &    1.9 &   28.7& 23.8  &  17.7      \\
0238+166  &  AO 0235+164        &      0.940  & 12.83  &  27.24$^{c}$ &     27.35   &   31.40       &   9 &   10.8 &    0.0&  0.0  &   5.3      \\
0301+346  &  MS 0257.9+3429     &      0.245  & 13.10  &  24.49$^{c}$ &     24.14   &    1.38       &   5 &    1.9 &   29.1& 24.1  &  17.8     \\
0314+247  &  RXS J0314.0+2445   &      0.054  & 12.71  &  22.95$^{d}$ &     22.60   &    ...        &   2 &    1.0 &   46.0& 36.0  &  25.7      \\
0326+024  &  2E 0323+0214       &      0.147  & 19.61  &  24.16$^{c}$ &     24.01   &    2.00       &  10 &    2.1 &   27.2& 22.8  &  17.0      \\
0416+010  &  2E 0414+0057       &      0.287  & 20.49  &  25.70$^{d}$ &     24.99   &    ...        &   1 &    2.1 &   26.6& 22.4  &  16.7      \\
0422+198  &  MS 0419.3+1943     &      0.512  & 16.63  &  25.15$^{c}$ &     24.73   &    1.14       &   5 &    2.3 &   24.7& 21.1  &  15.9      \\
0424+006  &  PKS 0422+004       &      0.310  & 15.01  &  26.33$^{b}$ &     26.32   &    ...        &   9 &    6.2 &    0.0&  8.9  &   8.5      \\
0505+042  &  RXS J0505.5+0416   &      0.027  & 16.87  &  23.57$^{d}$ &     23.17   &    ...        &   2 &    1.2 &   39.9& 31.7  &  22.8      \\
0507+676  &  1ES 0502+675       &      0.314  & 18.55  &  24.20$^{c}$ &     24.90   &    0.40       &   3 &    5.6 &    3.8& 10.2  &   9.2      \\
0508+845  &  S5 0454+84         &      0.112  & 12.59  &  23.75$^{c}$ &     25.19   &    5.10$^{f}$ &  4,9&   10.8 &    0.0&  0.0  &   5.3      \\
0509-040  &  4U 0506-03         &      0.304  & 17.77  &  25.57$^{c}$ &     24.82   &   36.00$^{f}$ &  11 &    1.9 &   28.7& 23.8  &  17.7      \\
0613+711  &  MS 0607.9+7108     &      0.267  & 14.41  &  24.03$^{c}$ &     24.39   &    0.40       &   5 &    3.5 &   16.4& 15.8  &  12.5     \\
0625+446  &  87GB 06216+4441    &      0.473* & 13.04  &  26.25$^{d}$ &     26.15   &     ...       &   8 &    5.5 &    5.2& 10.5  &   9.4      \\
0650+250  &  1ES 0647+250       &      0.203  & 17.85  &  24.85$^{c}$ &     24.83   &    4.89       &   3 &    3.2 &   17.9& 16.7  &  13.1      \\
0654+427  &  B3 0651+428        &      0.126  & 14.79  &  25.07$^{a}$ &     24.70   &    ...        &   2 &    2.4 &   24.2& 20.8  &  15.7      \\
0656+426  &  NPM1G +42.0131     &      0.059  & 17.34  &  25.85$^{c}$ &     24.30   &  699.00$^{f}$ &  12 &    0.9 &   49.5& 38.4  &  27.3      \\
0710+591  &  EXO 0706.1+5913    &      0.125  & 20.83  &  25.03$^{b}$ &     24.46   &    ...        &   7 &    1.9 &   29.3& 24.3  &  17.9      \\
0721+713  &  S5 0716+714        &      0.300  & 14.07  &  26.50$^{c}$ &     25.85   &  316.00$^{f}$ &   4 &    3.2 &   18.0& 16.8  &  13.1     \\
0738+177  &  PKS 0735+17        &      0.424  & 12.64  &  25.43$^{c}$ &     27.10   &   12.00$^{f}$ &   4 &   29.4 &    0.0&  0.0  &   0.0      \\
0744+745  &  MS 0737.9+7441     &      0.315  & 13.24  &  24.75$^{c}$ &     24.71   &    1.42       &   5 &    3.1 &   19.1& 17.5  &  13.6      \\
0753+538  &  S4 0749+54         &      0.200  & 12.23  &  25.47$^{e}$ &     26.12   &    ...        &   13&    9.3 &    0.0&  3.1  &   6.2      \\
0757+099  &  PKS 0754+100       &      0.266  & 12.48  &  25.26$^{c}$ &     26.56   &    6.70       &   9 &   17.7 &    0.0&  0.0  &   2.0      \\
0806+595  &  SBS 0802+596       &      0.300  & 16.43  &  25.26$^{e}$ &     24.81   &    ...        &   2 &    2.4 &   24.3& 20.9  &  15.7      \\
0809+523  &  1ES 0806+524       &      0.137  & 16.09  &  24.90$^{e}$ &     24.88   &    ...        &   3 &    3.3 &   17.4& 16.4  &  12.9      \\
0818+423  &  OJ 425             &      0.245  & 12.73  &  26.41$^{a}$ &     26.17   &    ...        &  14 &    5.0 &    8.3& 11.6  &  10.0      \\
0823+223  &  4C 22.21           &      0.951  & 12.94  &  28.53$^{c}$ &     26.96   &   602.60      &   4 &    2.7 &   21.4& 18.9  &  14.5      \\
0825+031  &  PKS 0823+033       &      0.506  & 11.79  &  25.23$^{c}$ &     26.98   &    1.40       &  14 &   29.4 &    0.0&  0.0  &   0.0      \\
0831+044  &  PKS 0829+046       &      0.174  & 12.81  &  25.74$^{b}$ &     25.95   &    ...        &   7 &    6.2 &    0.0&  9.0  &   8.6      \\
0831+087  &  1H 0827+089        &      0.941  & 13.90  &  26.67$^{d}$ &     26.15   &    ...        &  11 &    4.0 &   13.3& 14.0  &  11.4      \\
0832+492  &  OJ 448             &      0.548  & 12.33  &  25.98$^{c}$ &     26.36   &   23.40$^{f}$ &   4 &    8.4 &    0.0&  4.9  &   6.8      \\
0854+441  &  US 1889            &      0.382  & 17.31  &  26.05$^{a}$ &     25.06   &    ...        &   2 &    1.8 &   30.3& 24.9  &  18.4      \\
0854+201  &  OJ 287             &      0.306  & 12.75  &  25.25$^{c}$ &     26.56   &   17.00$^{f}$ &  4,9&   17.9 &    0.0&  0.0  &   2.0      \\
0915+295  &  B2 0912+29         &      0.302* & 15.64  &  26.20$^{b}$ &     25.59   &    ...        &   2 &    3.0 &   19.7& 17.9  &  13.8   \\
\hline
0916+526  &  RXS J0916.8+5238   &      0.271* & 17.01  &  25.61$^{d}$ &     24.92   &     ...       &   2 &    2.1 &   27.0& 22.7  &  16.9      \\
0929+502  &  RXS J0929.2+5013   &      0.370  & 13.76  &  26.08$^{e}$ &     26.50   &    ...        &   8 &    9.2 &    0.0&  3.1  &   6.2      \\
0930+498  &  1ES 0927+500       &      0.188  & 20.77  &  24.06$^{e}$ &     24.18   &    ...        &   3&     2.7 &   21.6& 19.1  &  14.6      \\
0930+350  &  B2 0927+35         &      0.302* & 14.35  &  26.48$^{b}$ &     25.95   &     ...       &   8 &    3.7 &   15.3& 15.2  &  12.1      \\
0952+656  &  RGB J0952+656      &      0.302* & 14.90  &  25.46$^{e}$ &     24.78   &     ...       &   2 &    2.0 &   28.0& 23.3  &  17.3      \\
0954+492  &  MS 0950.9+4929     &      0.380  & 16.73  &  24.20$^{c}$ &     24.06   &    0.26       &   5 &    2.1 &   26.6& 22.4  &  16.7      \\
0958+655  &  S4 0954+65         &      0.368  & 13.06  &  25.66$^{c}$ &     25.97   &   28.60$^{f}$ & 4,15&    6.8 &    0.0&  7.9  &   8.0      \\
1012+424  &  RXS J1012.7+4229   &      0.364  & 20.82  &  25.83$^{a}$ &     24.98   &    ...        &   2 &    1.9 &   28.7& 23.9  &  17.7      \\
1015+494  &  GB 1011+496        &      0.200  & 16.41  &  25.80$^{b}$ &     25.22   &    ...        &   16&    2.6 &   22.5& 19.7  &  14.9      \\
1031+508  &  1ES 1028+511       &      0.360  & 18.16  &  25.32$^{e}$ &     25.16   &    ...        &   3 &    3.4 &   17.0& 16.2  &  12.7      \\
1037+571  &  RXS J1037.7+5711   &      0.302* & 14.50  &  25.43$^{d}$ &     25.30   &     ...       &   2 &    3.7 &   15.2& 15.1  &  12.1      \\
1047+546  &  1ES 1044+549       &      0.473* & 12.86  &  24.68$^{d}$ &     24.32   &     ...       &   3 &    2.0 &   27.4& 22.9  &  17.1      \\
1053+494  &  MS 1050.7+4946     &      0.140  & 14.95  &  24.30$^{c}$ &     24.26   &    3.12       &   5 &    2.5 &   23.3& 20.2  &  15.3      \\
1104+382  &  MRK 421            &      0.031  & 18.20  &  24.40$^{a}$ &     24.14   &    ...        &   7&     2.0 &   27.7& 23.2  &  17.2      \\
1109+241  &  1ES 1106+244       &      0.271* & 16.70  &  24.96$^{d}$ &     24.50   &     ...       &   3 &    2.1 &   27.4& 22.9  &  17.1      \\
1120+422  &  EXO 1118.0+4228    &      0.124  & 17.21  &  24.10$^{d}$ &     23.83   &    ...        &   2 &    1.8 &   30.7& 25.2  &  18.6      \\
1136+701  &  MRK 180            &      0.046  & 18.73  &  25.16$^{c}$ &     23.80   &  237.00       &   3 &    0.8 &   52.2& 40.3  &  28.6      \\
1136+676  &  RXS J1136.5+6737   &      0.134  & 17.18  &  24.17$^{e}$ &     24.23   &    ...        &   2 &    2.6 &   22.0& 19.4  &  14.8      \\
1149+246  &  EXO 1449.9+2455    &      0.402  & 19.63  &  25.32$^{d}$ &     24.79   &    ...        &   2 &    2.2 &   25.8& 21.9  &  16.4      \\
1150+242  &  B2 1147+245        &      0.200  & 13.18  &  25.30$^{c}$ &     25.79   &   50.00$^{f}$ & 4,15&    7.1 &    0.0&  7.3  &   7.7     \\
1151+589  &  RXS J1151.4+5859   &      0.302* & 16.10  &  25.98$^{b}$ &     25.33   &     ...       &   2 &    2.6 &   22.6& 19.7  &  15.0      \\
1209+413  &  B3 1206+416        &      0.302* & 13.88  &  25.64$^{a}$ &     25.95   &     ...       &  8  &    6.7 &    0.0&  8.0  &   8.1      \\
1215+075  &  1ES 1212+078       &      0.130  & 15.57  &  24.80$^{d}$ &     24.56   &    ...        &   7 &    2.5 &   23.6& 20.4  &  15.4      \\
1217+301  &  B2 1215+30         &      0.130  & 15.10  &  25.46$^{b}$ &     25.24   &    ...        &   7 &    3.4 &   17.0& 16.2  &  12.7      \\
1220+345  &  GB2 1217+348       &      0.130  & 13.97  &  25.04$^{e}$ &     25.01   &    ...        &   2 &    3.5 &   16.5& 15.9  &  12.5      \\
1221+301  &  PG 1218+304        &      0.182  & 18.80  &  24.86$^{e}$ &     24.65   &    ...        &   7 &    2.6 &   22.4& 19.6  &  14.9      \\
1221+282  &  ON 231             &      0.102  & 14.16  &  25.12$^{c}$ &     25.43   &   41.00       &   3 &    5.3 &    6.2& 10.8  &   9.5      \\
1223+806  &  S5 1221+80         &      0.473* & 13.74  &  26.99$^{e}$ &     26.41   &     ...       &   8 &    4.3 &   11.8& 13.2  &  10.9      \\
1224+246  &  MS 1221.8+2452     &      0.218  & 13.60  &  24.25$^{c}$ &     24.38   &    1.05       &   5 &    3.0 &   19.7& 17.9  &  13.8      \\
1230+253  &  RXS J1230.2+2517   &      0.135  & 14.40  &  25.26$^{b}$ &     25.18   &    ...        &   2 &    3.6 &   15.8& 15.4  &  12.3      \\
1231+642  &  MS 1229.2+6430     &      0.164  & 15.84  &  24.33$^{c}$ &     24.43   &    2.34       &   5 &    3.0 &   19.7& 17.9  &  13.8      \\
1237+629  &  MS 1235.4+6315     &      0.297  & 15.61  &  24.38$^{c}$ &     24.48   &    0.70       &   17&    3.0 &   19.3& 17.6  &  13.6      \\
1241+066  &  1ES 1239+069       &      0.150  & 17.05  &  23.48$^{c}$ &     23.73   &    0.40       &   3 &    2.5 &   23.6& 20.4  &  15.4      \\
1248+583  &  PG 1246+586        &      0.847  & 14.27  &  26.90$^{b}$ &     27.17   &    ...        &   8 &   11.1 &    0.0&  0.0  &   5.1      \\
1253+530  &  S4 1250+53         &      0.302* & 14.39  &  26.45$^{b}$ &     25.89   &     ...       &   8 &    3.5 &   16.2& 15.7  &  12.4      \\
1257+242  &  1ES 1255+244       &      0.141  & 16.83  &  24.00$^{d}$ &     23.51   &    ...        &   3 &    1.3 &   38.0& 30.3  &  22.0      \\
1310+325  &  AUCVn              &      0.996  & 12.61  &  26.54$^{c}$ &     27.74   &    5.40       & 9,15&   27.7 &    0.0&  0.0  &   0.0     \\
1322+081  &  1ES 1320+084N      &      0.049  & 12.99  &  22.74$^{d}$ &     22.81   &    ...        &    3&    1.4 &   35.6& 28.6  &  20.8      \\
1341+399  &  RXS J1341.0+3959   &      0.163  & 19.97  &  25.19$^{a}$ &     24.33   &    ...        &    2&    1.4 &   35.5& 28.5  &  20.8      \\
1402+159  &  MC 1400+162        &      0.244  & 16.29  &  26.54$^{a}$ &     25.41   &    ...        &    1&    1.9 &   29.0& 24.1  &  17.8      \\
1404+040  &  MS 1402.3+0416(went&      0.344  & 15.74  &  25.76$^{c}$ &     24.80   &   11.89       &    3&    1.6 &   32.3& 26.4  &  19.3     \\
1409+596  &  MS 1407.9+5954     &      0.496  & 16.40  &  25.50$^{c}$ &     25.02   &    2.75       &    5&    2.6 &   22.8& 19.8  &  15.1      \\
1415+485  &  RGB J1415+485      &      0.500  & 14.13  &  25.73$^{e}$ &     25.57   &    ...        &    2&    4.1 &   13.1& 14.0  &  11.4      \\
1415+133  &  PKS 1413+135       &      0.247  & 12.39  &  26.66$^{a}$ &     26.01   &    ...        &    1&    3.5 &   16.5& 15.9  &  12.6      \\
1417+257  &  2E 1415+2557       &      0.237  & 19.00  &  25.29$^{d}$ &     24.74   &    ...        &    2&    2.1 &   26.5& 22.4  &  16.7      \\
1419+543  &  OQ 530             &      0.152  & 13.17  &  24.68$^{c}$ &     25.81   &   22.00$^{f}$ &  4,7&   11.4 &    0.0&  0.0  &   5.0      \\
1427+238  &  PKS 1424+240       &      0.160  & 15.34  &  25.69$^{a}$ &     25.18   &    ...        &    2&    2.7 &   21.9& 19.3  &  14.7      \\
1427+541  &  RGB J1427+541      &      0.106  & 14.79  &  24.38$^{e}$ &     23.80   &    ...        &    2&    1.4 &   36.4& 29.2  &  21.2      \\
1428+426  &  H 1426+428         &      0.129  & 18.41  &  24.43$^{e}$ &     23.93   &    ...        &    7&    1.6 &   33.5& 27.2  &  19.9      \\
1439+395  &  PG 1437+398        &      0.344  & 16.56  &  25.98$^{a}$ &     25.05   &    ...        &    2&    1.9 &   29.4& 24.3  &  18.0      \\
1442+120  &  1ES 1440+122       &      0.162  & 16.20  &  24.80$^{d}$ &     24.41   &    ...        &    3&    2.1 &   27.3& 22.9  &  17.0      \\
1444+636  &  MS 1443.5+6349     &      0.298  & 16.94  &  24.31$^{c}$ &     23.93   &    0.59       &    5&    1.7 &   31.6& 25.9  &  19.0      \\
1448+361  &  RXS J1448.0+3608   &      0.271* & 16.43  &  25.03$^{d}$ &     24.72   &     ...       &    2&    2.5 &   23.1& 20.1  &  15.2      \\
1458+373  &  B3 1456+375        &      0.333  & 12.86  &  25.91$^{a}$ &     25.93   &    ...        &    8&    5.4 &    5.7& 10.7  &   9.5      \\
1501+226  &  MS 1458.8+2249     &      0.235  & 14.69  &  24.73$^{c}$ &     25.06   &    2.66       &    5&    4.6 &   10.4& 12.5  &  10.5     \\
\hline
1509+559  &  SBS 1508+561       &      0.302* & 14.83  &  24.83$^{e}$ &     24.80   &     ...       &    2&    3.2 &   18.3& 17.0  &  13.2      \\
1516+293  &  RXS J1516.7+2918   &      0.130  & 18.56  &  25.01$^{a}$ &     24.13   &    ...        &    2&    1.3 &   38.1& 30.4  &  22.0      \\
1517+654  &  1H 1515+660        &      0.702  & 17.82  &  25.72$^{e}$ &     25.39   &    ...        &    2&    3.3 &   17.4& 16.4  &  12.9      \\
1532+302  &  RXS J1532.0+3016   &      0.064  & 16.86  &  23.87$^{d}$ &     23.64   &    ...        &    2&    1.7 &   32.0& 26.1  &  19.2      \\
1533+342  &  RXS J1533.4+3416   &      0.810  & 17.89  &  25.78$^{e}$ &     25.75   &    ...        &    2&    4.9 &    8.9& 11.9  &  10.1      \\
1534+372  &  RGB J1534+372      &      0.143  & 13.97  &  24.02$^{e}$ &     23.98   &    ...        &    2&    2.2 &   25.8& 21.8  &  16.3      \\
1535+533  &  1ES 1533+535       &      0.890  & 19.55  &  25.96$^{d}$ &     25.31   &    ...        &    3&    2.5 &   22.8& 19.9  &  15.1      \\
1536+016  &  MS 1534.2+0148     &      0.312  & 18.67  &  25.53$^{c}$ &     24.79   &    8.78       &    5&    1.9 &   29.1& 24.1  &  17.8      \\
1540+819  &  1ES 1544+820       &      0.271* & 17.53  &  25.42$^{e}$ &     24.89   &     ...       &    3&    2.3 &   24.9& 21.3  &  16.0      \\
1540+147  &  4C 14.6            &      0.605  & 14.25  &  27.58$^{c}$ &     27.10   &  202.00       &    9&    6.3 &    0.0&  8.7  &   8.4      \\
1542+614  &  RXS J1542.9+6129   &      0.302* & 14.19  &  25.27$^{e}$ &     25.36   &    ...        &    2&    4.4 &   11.3& 13.0  &  10.8      \\
1554+201  &  MS 1552.1+2020     &      0.222  & 16.89  &  25.09$^{c}$ &     24.58   &    6.86       &    5&    2.1 &   27.3& 22.9  &  17.0     \\
1555+111  &  PG 1553+11         &      0.360  & 15.92  &  26.29$^{b}$ &     26.11   &    ...        &    2&    5.1 &    7.7& 11.4  &   9.9      \\
1602+308  &  RXS J1602.2+3050   &      0.302* & 16.27  &  25.36$^{d}$ &     24.65   &     ...       &    2&    1.8 &   29.6& 24.5  &  18.1      \\
1626+352  &  RXS J1626.4+3513   &      0.497  & 15.05  &  25.39$^{d}$ &     24.95   &    ...        &    2&    2.5 &   23.0& 20.0  &  15.1      \\
1644+457  &  RXS J1644.2+4546   &      0.225  & 17.28  &  25.67$^{a}$ &     24.89   &    ...        &    2&    1.9 &   28.4& 23.7  &  17.5      \\
1652+403  &  RGB J1652+403      &      0.240  & 14.80  &  24.60$^{d}$ &     24.19   &    ...        &    2&    1.8 &   29.6& 24.5  &  18.1      \\
1653+397  &  MRK 501            &      0.034  & 16.17  &  23.79$^{c}$ &     24.51   &   67.00$^{f}$ &  4,7&    4.8 &    9.3& 12.0  &  10.2      \\
1704+716  &  RXS J1704.8+7138   &      0.350  & 15.32  &  25.06$^{e}$ &     24.72   &    ...        &   2 &    2.5 &   23.6& 20.4  &  15.4      \\
1719+177  &  PKS 1717+177       &      0.137  & 12.43  &  25.19$^{a}$ &     25.42   &    ...        &   2 &    5.0 &    7.9& 11.5  &   9.9      \\
1724+400  &  B2 1722+40         &      1.049  & 12.64  &  27.81$^{a}$ &     26.99   &    ...        &   8 &    4.7 &    9.7& 12.2  &  10.3      \\
1725+118  &  H 1722+119         &      0.018  & 15.75  &  23.06$^{d}$ &     22.80   &    ...        &   2 &    1.1 &   41.6& 32.9  &  23.6      \\
1728+502  &  IZw187             &      0.055  & 17.16  &  24.20$^{b}$ &     23.96   &    ...        &  10 &    1.9 &   28.9& 24.0  &  17.8      \\
1739+476  &  OT 465             &      0.473* & 13.34  &  27.13$^{b}$ &     26.69   &     ...       &   8 &    5.4 &    5.5& 10.6  &   9.4      \\
1742+597  &  RGBJ 1742+597      &      0.400  & 13.75  &  25.72$^{e}$ &     25.49   &    ...        &   2 &    3.7 &   15.0& 15.0  &  12.0      \\
1743+195  &  NPM1G +19.0510     &      0.084  & 17.44  &  24.40$^{c}$ &     24.53   &   11.60       &   3 &    3.2 &   18.2& 17.0  &  13.2      \\
1745+398  &  B3 1743+398B       &      0.267  & 17.59  &  26.55$^{a}$ &     25.31   &    ...        &   2 &    1.7 &   31.6& 25.9  &  19.0      \\
1747+469  &  B3 1746+470        &      1.484  & 13.30  &  27.26$^{a}$ &     27.41   &    ...        &   8 &   11.3 &    0.0&  0.0  &   5.0      \\
1748+700  &  S4 1749+70         &      0.770  & 13.80  &  26.63$^{c}$ &     26.91   &   45.70$^{f}$ & 4,14&    9.9 &    0.0&  0.0  &   5.8      \\
1749+433  &  B3 1747+433        &      0.473* & 13.08  &  26.62$^{a}$ &     26.21   &    ...        &   8 &    4.5 &   11.1& 12.9  &  10.7      \\
1750+470  &  RXS J1750.0+4700   &      0.160  & 18.31  &  25.28$^{a}$ &     23.78   &    ...        &   2 &    0.7 &   55.5& 42.7  &  30.2      \\
1751+096  &  PKS 1749+096       &      0.322  & 11.33  &  24.92$^{c}$ &     26.99   &    2.00       &  14 &   37.1 &    0.0&  0.0  &   0.0     \\
1756+553  &  RXS J1756.2+5522   &      0.271* & 19.74  &  24.72$^{e}$ &     24.26   &     ...       &    2&    1.8 &   29.7& 24.6  &  18.1      \\
1757+705  &  MS 1757.7+7034     &      0.407  & 13.37  &  24.65$^{c}$ &     24.64   &    0.62       &    5&    3.0 &   19.3& 17.6  &  13.6      \\
1800+784  &  S5 1803+784        &      0.680  & 13.35  &  27.57$^{e}$ &     27.36   &    ...        &4,15 &    8.5 &    0.0&  4.7  &   6.7      \\
1806+698  &  3C 371             &      0.051  & 14.42  &  25.87$^{c}$ &     24.95   &   990.00      & 4,7 &    1.8 &   30.4& 25.0  &  18.4      \\
1808+468  &  RGB J1808+468      &      0.450  & 14.34  &  25.85$^{e}$ &     25.33   &    ...        &   2 &    2.8 &   20.8& 18.6  &  14.3      \\
1811+442  &  RGB J1811+442      &      0.350  & 15.30  &  25.92$^{e}$ &     24.26   &    ...        &   2 &    0.8 &   52.3& 40.4  &  28.7      \\
1813+317  &  B2 1811+31         &      0.117  & 15.34  &  25.00$^{b}$ &     24.37   &    ...        &   2 &    1.7 &   31.1& 25.5  &  18.8      \\
1824+568  &  4C 56.27           &      0.664  & 12.56  &  28.02$^{c}$ &     26.99   &  452.00       &   9 &    4.1 &   13.2& 14.0  &  11.4      \\
1829+540  &  RXS J1829.4+5402   &      0.302* & 15.06  &  25.73$^{e}$ &     24.61   &     ...       &   2 &    1.3 &   37.1& 29.7  &  21.5      \\
1838+480  &  RXS J1838.7+4802   &      0.300  & 13.24  &  25.05$^{d}$ &     24.71   &    ...        &   2 &    2.5 &   23.6& 20.4  &  15.4      \\
1841+591  &  RGB J1841+591      &      0.530  & 14.94  &  25.68$^{e}$ &     24.64   &    ...        &   2 &    1.4 &   35.4& 28.5  &  20.7      \\
1853+672  &  1ES 1853+671       &      0.212  & 16.25  &  23.81$^{c}$ &     24.12   &    0.40       &   3 &    3.0 &   19.5& 17.7  &  13.7      \\
1927+612  &  S4 1926+61         &      0.473* & 12.72  &  26.62$^{b}$ &     26.67   &     ...       &  8  &    7.7 &    0.0&  6.3  &   7.3      \\
1959+651  &  1ES 1959+650       &      0.047  & 17.33  &  23.01$^{c}$ &     24.10   &    1.60       &  3  &    5.2 &    6.9& 11.1  &   9.7      \\
2005+778  &  S5 2007+77         &      0.342  & 12.39  &  26.14$^{c}$ &     26.38   &   28.90       &  9  &    7.7 &    0.0&  6.2  &   7.3      \\
2009+724  &  S5 2010+72         &      0.473* & 12.94  &  27.02$^{e}$ &     26.90   &     ...       &  8  &    7.5 &    0.0&  6.7  &   7.5      \\
2022+761  &  S5 2023+76         &      0.473* & 13.54  &  26.65$^{e}$ &     26.39   &     ...       &  13 &    5.4 &    5.7& 10.7  &   9.5      \\
2039+523  &  1ES 2037+521       &      0.053  & 15.60  &  22.27$^{c}$ &     23.31   &    0.23       & 3   &    3.6 &   16.0& 15.6  &  12.4      \\
2134-018  &  PKS 2131-021       &      1.285  & 12.04  &  27.93$^{c}$ &     27.87   &   70.80       &  14 &   11.9 &    0.0&  0.0  &   4.7      \\
2145+073  &  MS 2143.4+0704     &      0.237  & 13.61  &  25.13$^{c}$ &     24.79   &    6.60       &   5 &    2.5 &   22.9& 20.0  &  15.1      \\
2152+175  &  PKS 2149+17        &      0.473* & 13.00  &  26.03$^{c}$ &     26.57   &    36.70$^{f}$&   4 &   10.4 &    0.0&  0.0  &   5.5      \\
2202+422  &  BL LAC             &      0.070  & 13.15  &  24.21$^{c}$ &     25.73   &   40.00$^{f}$ & 4,7 &   14.5 &    0.0&  0.0  &   3.5      \\
2250+384  &  B3 2247+381        &      0.119  & 15.35  &  24.64$^{a}$ &     24.29   &    ...        &  2  &    2.0 &   27.6& 23.1  &  17.2      \\
\hline
2257+077  &  PKS 2254+074       &      0.190  & 13.31  &  24.78$^{c}$ &     25.66   &   17.00$^{f}$ &9,10 &    8.9 &    0.0&  4.0  &   6.4      \\
2319+161  &  Q J2319+161        &      0.302* & 15.29  &  25.13$^{d}$ &     24.58   &     ...       &    2&    2.0 &   27.9& 23.3  &  17.3      \\
2322+346  &  TEX 2320+343       &      0.098  & 16.68  &  24.59$^{b}$ &     23.82   &   ...         &    2&    1.2 &   39.4& 31.3  &  22.6      \\
2323+421  &  1ES 2321+419       &      0.059  & 13.05  &  23.09$^{d}$ &     23.18   &    ...        &   11&    1.7 &   31.5& 25.8  &  18.9      \\
2329+177  &  1ES 2326+174       &      0.213  & 17.84  &  24.64$^{d}$ &     24.30   &    ...        &   3 &    2.0 &   27.4& 23.0  &  17.1      \\
2339+055  &  MS 2336.5+0517     &      0.740  & 14.89  &  25.74$^{d}$ &     24.88   &    ...        &   5 &    1.8 &   30.0& 24.7  &  18.2     \\
2347+517  &  1ES 2344+514       &      0.044  & 15.86  &  23.30$^{c}$ &     23.97   &    3.58       &   3 &    3.6 &   15.6& 15.3  &  12.2      \\
2350+196  &  MS 2347.4+1924     &      0.515  & 15.72  &  24.80$^{d}$ &     24.29   &    ...        &   5 &    1.8 &   30.0& 24.8  &  18.3      \\
\hline
\end{tabular}
\smallskip
 Notes: Col.~1: the source IAU name (J2000). Col.~2:
the source alias name. Col.~3: the redshift. `*' indicate that the
redshift is unknown, and taken as the average redshift of
LBL/IBL/HBL subclass. Col.~4: the intrinsic synchrotron peak
frequency after excluding the Doppler factor. Col.~5: the total
radio power at 408~MHz (notes: `a' -- from 408~MHz; `b' -- from 360~MHz; `c' -- from extended flux; `d' -- from total 1.4~GHz; `e' -- from 330~MHz). Col.~6: the 5~GHz core luminosity. Col.~7: the 5~GHz extended flux density in mJy, `f' indicates the extend flux at 1.4~GHz. Col.~8: the references for core and extended flux: (1: \cite{liuandzhang}  2: \cite{laure99}  3:    \cite{per96} 4:     \cite{rec01}  5:    \cite{cavallotti04}  6:  \cite{cas99}  7:     \cite{gir04b}  8:     \cite{Tay96}  9:    \cite{mur93}  10:     \cite{lau93}  11:     \cite{koll}  12:    \cite{rec03} 13:  \cite{taylor94}  14:    \cite{re03}  15:    \cite{cas02}  16:     \cite{AWB98} 17:     \cite{rec00}). Col.~9: the Doppler factor. Cols.~10--12: the viewing angle of $\Gamma= 3$, 5 and~10, respectively.
\end{table}